Author's response to reviews


Authors:

Godfrey S Mfinanga Dr (gsmfinanga@yahoo.com)
Godfather D Kimaro Dr (fatherdk2002@yahoo.com)
Esther Ngadaya Dr (engadaya@yahoo.com)
Sirili Massawe Prof (smassawe@muchs.ac.tz)
Rugola Mtandu Dr (rugorugo@yahoo.com)
Elizabeth E Shayo Mrs (bshayo@yahoo.com)
Amos Kahwa Dr (akahwa@hotmail.com)
Ominde Achala Dr (oachala@crhcs.or.tz)
Alice Mutungi Dr (amutungi@rcqhc.org)
Stanton Cynthia Dr (cstanton@jhsph.edu)
Deborah Armbruster Ms (darmbruster@path-dc.org)
Anderw Kitua Dr (akitua@nimr.or.tz)
David Sintasath Dr (dsintasa@jhsph.edu)
Rod Knight Dr (rodjknight@aol.com)

Version: 3 Date: 19 February 2009

Author's response to reviews:

Dear Harps Editorial team,

We have revised the manuscript as per reviewers' comments. Our responses inserted in the manuscript are in blue color.

Hereunder find our point-by-point responses of the comments form each of the two reviewers.

Reviewer 1:

The reviewer had no additional comment as s/he have accepted the paper.

Reviewer 2:

1. Comment: In the discussion section ergometrine is incorrectly spelled as
ergomentrine in a couple of instances. This need to be checked and changed.

Response: spelling has been checked and changed.

2. Comment: In the second paragraph of the discussion section the last sentence is confusing and appears to be referring to this study rather than citing another study. It should be reworded.

Response: The sentence has been reworded.

3. Comment: In the discussion paragraph that begins “the U.S Pharmacopeia…”, there is a sentence that begins , the facility that stored oxytocin …”. This sentence is very confusing and should be reworded.

Response: The sentence has been reworded.

4. Comment: In conclusion section, the sentence that begins “The drug for AMTSL…” “was” should be replaced with “were”

Response: The replacement has been done.

5. Comment: Table two needs to be reformatted to not split the column headings. Also I do not understand why there is a number below numbers in the column for Ns in the same table.

Response: The table has been reformatted as suggested.